Author's response to reviews

Title: Birth characteristics and the risk of childhood leukaemias and lymphomas in New Zealand: a case-control study.

Authors:

Donny IM Wong (wondo585@student.otago.ac.nz)
John D Dockerty (john.dockerty@stonebow.otago.ac.nz)

Version: 4 Date: 14 September 2006

Author's response to reviews:

14 September, 2006

To The BioMed Central Editorial Production Team,

Re: Birth characteristics and the risk of childhood leukaemias and lymphomas in New Zealand: a case-control study.

We have made the formatting changes requested and checked our manuscript before resubmission. We are satisfied with it. Many thanks for your assistance.

Yours sincerely,

Donny IM Wong and John D Dockerty